Author's response to reviews

Title: Severe hydrops in an infant of a Rhesus D-positive mother due to anti-c antibodies diagnosed antenatally: a case report

Authors:

Shilpa Singla (shilpasingla1@gmail.com)
Sunesh Kumar (suneshkumar@hotmail.com)
Kallol Kumar Roy (drkkroy2003@yahoo.co.in)
Jai Bhagwan Sharma (jbsharma2000@gmail.com)
Garima Kachhawa (garimakhadagawat@hotmail.com)

Version: 7 Date: 1 February 2010

Author's response to reviews:

To
The Editor
JMCR

This is regarding manuscript number ID 4836114782272454 which has been editorially accepted by your esteemed journal. The Author names in the manuscript have been corrected. The abstract as suggested now includes the ethnicity of the patient. The abstract and the manuscript have been checked.

Kindly proceed with the publication of the paper.

Regards
Dr Shilpa Singla